Author's response to reviews

Title: Deep sternal wound infection after cardiac surgery

Authors:

Hiroshi Kubota (kub@ks.kyorin-u.ac.jp)
Hiroaki Miyata (hiroaki.miyata@gmail.com)
Noboru Motomura (nmotomura@gmail.com)
Minoru Ono (minoruono61@hotmail.com)
Shinichi Takamoto (takamoto-cvs@umin.ac.jp)
Kiyonori Harii (harii@ks.kyorin-u.ac.jp)
Norihiro Oura (norihiro.ohura@gmail.com)
Shinichi Hirabayashi (shinichi@med.teikyo-u.ac.jp)
Shunei Kyo (kyo_shunei@mail.goo.ne.jp)

Version: 2 Date: 8 May 2013

Author's response to reviews:

Editors-in-Hief
Mr. Vipin Zamvar
Journal of Cardiothoracic Surgery
Dear, Mr. Zamvar
Dear, Editors

Thank you for giving us an opportunity to re-submit our manuscript. We are sorry to be late to re-submit it. We re-write the manuscript along with reviewers' comments. Partially, according to the Database limitation, we could not answer to the reviewer's comments. We responde to the reviewer the reason why we could not answer fully. We added many new statistical data according to the reviewer's comment. I deeply appreciate to the editors and reviewers who gave us an opportunity to progress our manuscript. I feel sorry for my poor English.

Sincerely,
Hiroshi KUBOTA M.D.
Kyorin Univ. Tokyo, Japan